Author’s response to reviews

Title: Handling informative dropout in longitudinal analysis of health-related quality of life: Application of three approaches to data from the esophageal cancer clinical trial PRODIGE 5/ACCORD 17

Authors:

Benjamin Cuer (benjamin.cuer@icm.unicancer.fr; cuerbenjamin@gmail.com)

Caroline Mollevi (Caroline.mollevi@icm.unicancer.fr)

Amélie Anota (aanota@chu-besancon.fr)

Emilie Charton (echarton@chu-besancon.fr)

Beata Juzyna (b-juzyna@unicancer.fr)

Thierry Conroy (t.conroy@nancy.unicancer.fr)

Célia Touraine (Celia.Touraine@icm.unicancer.fr)

Version: 3 Date: 19 Aug 2020

Author’s response to reviews:

We have updated the section "Authors contributions" in the manuscript. The two authors BJ and TC participated to the acquisition of data. Then, we have clarified their contribution (Declaration section, line 531, page 24) as follows: “BJ and TC participated in the day-to-day running of the trial and contributed to the acquisition of the data”.